**Author’s response to reviews**

**Title:** The experience of palliative patients and their families of a family meeting utilised as an instrument for spiritual and psychosocial care: A qualitative study

**Authors:**

- Heather M Tan (Heather.Tan@monash.edu)
- Anne Wilson (anne.wilson@adelaide.edu.au)
- Ian Olver (ian.olver@cancer.org.au)
- Christopher Barton (christopher.barton@adelaide.edu.au)

**Version:** 8 **Date:** 11 March 2011

**Author's response to reviews:** see over
Dear Editor

The reference sequence has been corrected. It is shown as track changes on pages 8 and 10. The actual reference list is correct as it was reflecting the changes which have now been made on the above pages.

I apologise for overlooking those two errors in the last version

Heather Tan